

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Agarwal, Vikas; Ma, Linlin; Mullally, Kevin

Working Paper Managerial multitasking in the mutual fund industry

CFR Working Paper, No. 13-10 [rev.]

Provided in Cooperation with: Centre for Financial Research (CFR), University of Cologne

Suggested Citation: Agarwal, Vikas; Ma, Linlin; Mullally, Kevin (2015) : Managerial multitasking in the mutual fund industry, CFR Working Paper, No. 13-10 [rev.], University of Cologne, Centre for Financial Research (CFR), Cologne

This Version is available at: https://hdl.handle.net/10419/114470

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Managerial Multitasking in the Mutual Fund Industry^{*}

Vikas Agarwal Linlin Ma Kevin Mullally

June 2015

^{*}Vikas Agarwal is from Georgia State University, J. Mack Robinson College of Business, 35 Broad Street, Suite 1234, Atlanta GA 30303, USA. Email: vagarwal@gsu.edu. Tel: +1-404-413-7326. Fax: +1-404-413-7312. Vikas Agarwal is also a Research Fellow at the Centre for Financial Research (CFR), University of Cologne. Linlin Ma is from D'Amore-McKim School of Business, Northeastern University, 360 Huntington Avenue, Boston MA 02115, USA. Email: l.ma@neu.edu. Tel: +1-617-373-4569. Kevin Mullally is from Georgia State University, J. Mack Robinson College of Business, 35 Broad Street, Suite 1229, Atlanta GA 30303. Email: kmullally1@gsu.edu. Tel: +1-404-413-7318. We are grateful to the following for their comments: Jonathan Berk, Sudheer Chava, Gjergji Cici, Chris Clifford, Naveen Daniel, Nishant Dass, Gerald Gay, Simon Gervais, Lixin Huang, Narayanan Jayaraman, Wei Jiang, Bradford Jordan, Jayant Kale, Jerchern Lin, Pedro Matos, Felix Meschke, Jeffrey Pontiff, Veronika Krepely Pool, David Rakowski, Jonathan Reuter, Chip Ryan, Mila Getmansky Sherman, Marta Szymanowska, Qinghai Wang, Lei Wedge, Russ Wermers and seminar and conference participants at the AFA 2011 Meetings, Georgia Institute of Technology, University of Kentucky, the 5th Conference on Professional Asset Management, the 5th Singapore International Conference, and the FMA 2011 Meetings. This paper won the Best Paper Award in Investments sponsored by the AAII at the FMA 2011 Meetings. We are thankful to Rong Shao for excellent research assistance, and Melissa Pugeda and Steven Arnold of Morningstar for assistance with the data. We are responsible for all errors.

Managerial Multitasking in the Mutual Fund Industry

Abstract

We examine the determinants and consequences of mutual fund managers simultaneously managing multiple funds. Well-performing managers multitask by taking over poorly performing funds or launching new funds. Subsequent to multitasking, funds run by managers prior to multitasking (*i.e.*, incumbent funds) experience performance deterioration while the performance of the acquired funds improves. Multitasking increases the assets of fund companies but results in a wealth transfer from shareholders of the incumbent funds to those of the funds the managers take over. Multitasking arrangements are terminated when investors recognize the associated agency problem and withdraw their capital from the incumbent funds.

Keywords: Multitasking, Fund Performance, Fund Flows, Agency Problems

JEL Classification: G10, G20, G23

It is commonly believed that mutual fund companies assign a single fund to a portfolio manager. For example, the Fidelity Magellan Fund was the only fund run by their star manager, Peter Lynch. In reality, fund companies frequently assign multiple funds to the same portfolio manager. For instance, Will Danoff, manager of Fidelity Contrafund since 1990, also began managing Fidelity New Insights Fund in 2003. Moreover, 48% of mutual fund managers managed multiple funds simultaneously (*i.e.*, multitask) and these managers controlled about 62% of the total assets in the industry during our sample period of 1980 to 2012. Despite being a widely prevalent practice, there has been little academic research on the subject of managerial multitasking in the mutual fund industry (Yadav, 2011; and Choi, Kahraman, and Mukherjee, 2014 being two exceptions). We attempt to fill this gap in the literature by examining the determinants and consequences of the multitasking phenomenon in the mutual fund industry.

We identify a sample of managers of U.S. open-end equity mutual funds that switch from single-tasking (*i.e.*, managing one fund, which we refer to as *incumbent*) to multitasking (*i.e.*, managing multiple funds) by either taking over existing funds within fund companies (which we refer to as *acquired*) or by launching *new* funds.¹ We refer to the acquired and new funds together as new-task funds. We document several findings that shed light on the economics of multitasking.

We find that managers who switch to multitasking exhibit superior past performance in the incumbent funds prior to the switch. Moreover, these managers multitask either by taking over other funds in the fund companies that are poorly performing or by launching new funds. We offer three explanations for these findings. First, well-performing managers of incumbent

¹ We borrow the terms, *incumbent* and *acquired*, from the mergers and acquisitions literature although our paper is not about mutual fund mergers, which have been studied by Jayaraman, Khorana, and Nelling (2002).

funds can generate a positive spillover effect in form of greater investor flows into the acquired funds and new funds. Second, multitasking mechanism can help fund companies to turn around poorly performing funds, whose presence can adversely affect companies' reputation. Lastly, since multitasking arrangement increases the manager's span of control, mutual fund companies can use it to retain their good managers and to replace their bad managers, thereby maximizing the economic surplus generated through their monitoring role.²

We next examine the implications of managerial multitasking for fund performance, for which we have three hypotheses. First, managers are likely to be more distracted when running multiple funds simultaneously. These distractions can negatively impact managers' performance. This *distraction hypothesis* predicts that both the performance of the incumbent and the acquired funds suffer when managers multitask. Second, multitasking managers can divert their attention and effort from the existing funds to the new task. This *effort diversion hypothesis* predicts performance deterioration for the incumbent funds but performance improvement for the acquired funds. Third, multitasking managers can exploit synergistic benefits associated with learning while conducting investment research for multiple funds. ³ This *synergy creation hypothesis* predicts that performance of both the incumbent and acquired funds improve after the managers' switch to multitasking.

To test these hypotheses, we compare the performance of the incumbent funds and the acquired funds before and after their managers' switch to multitasking. We find that there is a striking decline in the risk-adjusted performance of the incumbent funds over the 12-month period subsequent to the switch – a decline of 0.13, 1.77%, and 1.23% in the annualized Sharpe

² Gervais, Lynch, and Musto (2005) theoretically model mutual fund companies as delegated monitors of money managers, who can credibly convey manager quality and generate value through their firing and retention decisions. ³ This notion is similar to cross-learning documented in the context of firms offering hedge funds and funds of hedge funds simultaneously, as studied by Agarwal, Lu, and Ray (2015).

ratio, the Carhart (1997) four-factor alpha and the benchmark-adjusted alpha, respectively. In contrast, there is an improvement in the performance of the acquired funds: 0.21, 1.91%, and 2.89% using the annualized Sharpe ratio, four-factor alpha and benchmark-adjusted alpha, respectively. These changes in the three performance measures are economically significant since the average values for the incumbent funds prior to multitasking are 0.84, 1.39%, and 1.68%, while the corresponding values for the acquired funds are 0.46, -2.15%, and -1.93%. We interpret these results being consistent with the effort diversion hypothesis, and not in favor of either the distraction or the synergy creation hypothesis.

An obvious concern is that the above results may be driven by mean reversion in fund performance and/or decreasing returns to scale. To allay such concerns, we use propensity score matching to construct samples of *control* funds that are similar to our *treated* groups of incumbent and acquired funds but whose managers remain single-tasking. That is, the difference between the two groups allows us to capture the treatment effect of multitasking. We continue to find performance decline and improvement in the incumbent and acquired funds, respectively, even after taking into account the changes in performance of the respective matched control funds. In fact, unlike our treated samples of incumbent and acquired funds, the matched control samples show virtually no change in performance. Together, these findings suggest that mean reversion in fund performance and decreasing returns to scale do not entirely explain our results related to the effect of multitasking on fund performance.

To further corroborate the effort diversion hypothesis, we examine if the exertion of greater effort in the new-task fund is associated with more pronounced deterioration in the performance of the manager's incumbent fund. Since it is challenging to observe and measure effort, we use three proxies for effort in the new task that include managing a new fund versus an

existing fund, new-task funds' turnover ratio, and new-task funds' deviation from the Morningstar style index.⁴ Each of these proxies attempts to capture the intensity of multitasking managers' active involvement in the new-task funds. Consistent with our effort diversion hypothesis, we find that the incumbent funds' performance suffers more when their managers launch a new fund and manage funds with greater turnover ratio and larger style deviation.

Next, we examine the economic incentives of mutual fund companies to engage in these multitasking arrangements by analyzing their effect on investor flows. We compare the net dollar flows into the incumbent and acquired funds before and after their managers' switch to multitasking. We find that incumbent funds do not display a significant change in investor flows while acquired funds are rewarded with greater investor flows over the 12-month period subsequent to the switch. Further, new funds launched by multitasking managers attract more investor flows compared to the ones launched by single-tasking managers. These findings are consistent with the aforementioned positive spillover effect of multitasking on investor flows that allows mutual fund companies to increase their assets. Finally, in well-functioning financial markets with rational investors, fund companies should terminate these multitasking arrangements if investors withdraw their capital from the poorly performing incumbent funds. We find that this is indeed the case.

Taken together, our findings uncover an important and hitherto unexplored manifestation of potential agency problems in the form of managerial multitasking in the mutual fund industry. By assigning multiple funds to the same portfolio manager, fund companies benefit from managerial multitasking by increasing their assets, turning around their poorly performing funds,

⁴ The deviation from the style index is similar to the active share measure proposed by Cremers and Petajisto (2009) for mutual funds and strategy distinctiveness measure proposed by Sun, Wang, and Zheng (2012) for hedge funds.

and retaining their well-performing managers. However, investors of the incumbent funds bear the costs of poor performance due to managers' multitasking. We show that these costs largely accrue to investors in the short run as fund performance improves after value-destroying multitasking arrangements are discontinued. Our work thus contributes to the large literature on the agency problems in the delegated asset management industry.⁵

In addition, our paper complements the growing literature studying how fund performance relates to different organizational structures such as team management (*e.g.*, Bliss, Potter, and Schwarz, 2008; Massa, Reuter, and Zitzewitz, 2010; Baer, Kempf, and Ruenzi, 2011; and Patel and Sarkissian, 2014), side-by-side management (*e.g.*, Cici, Gibson, and Moussawi, 2010; Nohel, Wang, and Zheng, 2010; and Deuskar et al., 2011), and outsourcing arrangement (*e.g.*, Chen, Hong, and Kubik, 2013) in the mutual fund industry. Finally, our paper relates to the corporate finance literature that studies whether firms with directors serving multiple boards are associated with weak corporate governance (*e.g.*, Ferris, Jagannathan, and Pritchard, 2003; and Fich and Shivdasani, 2006).

The rest of our paper is organized as follows. Section I describes the data, sample selection, and construction of key variables. Section II examines the characteristics of funds associated with multitasking. Section III studies the performance implications of managerial multitasking. Section IV discusses the economic incentives of the fund companies to engage in

⁵ This literature includes the window-dressing behavior among portfolio managers (*e.g.*, Lakonishok et al., 1991; He, Ng, and Wang, 2004; Ng and Wang, 2004; Meier and Schaumburg, 2006;, and Agarwal, Gay, and Ling, 2014), strategic risk-shifting motivated by agency issues (*e.g.*, Brown, Harlow, and Starks, 1996; Chevalier and Ellison, 1997; Kempf and Ruenzi, 2008; Kempf, Ruenzi, and Thiele, 2009; Hu et al., 2011; Huang, Sialm, and Zhang, 2011; and Schwarz, 2012), conflict of interests arising from offering multiple products (*e.g.*, Gaspar, Massa, and Matos, 2006; Chen and Chen, 2009; Cici, Gibson, and Moussawi, 2010; Bhattacharya, Lee, and Pool, 2013; and Sandhya, 2012) and incentive misalignment due to business ties (*e.g.*, Davis and Kim, 2007; Cohen and Schmidt, 2009; and Ashraf, Jayaraman, and Ryan 2012).

multitasking by analyzing its effect on investor flows. Section V focuses on fund companies' decisions to terminate multitasking arrangements. Section VI concludes.

I. Data Sample and Construction of Variables

A. Data Sample

The primary data source for our study is the survivorship-bias free Morningstar Direct Mutual Fund database. This database covers U.S. open-end mutual funds and provides information about fund names, manager names, returns, assets, inception dates, expense ratios, turnover ratios, net dollar flows, investment styles, fund tickers, fund CUSIPs, and other fund characteristics. We rely primarily on the Morningstar database for two reasons. First, manager information is available over a longer time period in the Morningstar database than it is in the CRSP database.⁶ Second, manager information is more accurate in the Morningstar database than in the CRSP database (see Massa, Reuter, and Zitzewitz, 2010).

We focus only on actively managed equity funds that have more than 50% of their assets invested in common stocks and we exclude funds whose managers are anonymous. We also exclude team-managed funds since task allocation among different team members is not observable. This yields a final sample of 3,316 portfolio managers from 4,195 domestic equity funds, covering 268,205 fund-month observations between 1980 and 2012.

We construct time-series data on manager-fund pairs that allows us to identify the exact month in which the managers switch from single-tasking to multitasking by tracking the number of funds they manage over time. We use the month in which a manager switches from managing one fund (*i.e.*, single-tasking) to more than one fund (*i.e.*, multitasking) as the event month for

⁶ Morningstar provides information on mutual fund managers from 1949 onward while this information is only available in CRSP beginning in 1993.

our empirical analyses. To avoid the cases of temporary arrangements, we require the managers to (a) have at least 12-month tenure in the incumbent funds before switching to multitasking, and (b) continue managing both the incumbent fund and the new-task fund (i.e., acquired fund or new fund) for at least 12 months after the switch. Using this criterion, we find a total of 688 managers that switch from single-tasking to multitasking: 274 cases (39.83%) where the managers acquire an existing fund, 315 cases (45.78%) where the managers launch a new fund, and 99 cases (14.39%) where the managers is entrusted with more than one new-task fund.⁷ The sample of new-task funds consists of 298 acquired funds and 236 new funds. As for the control group for multitasking managers, we find 82,611 fund-month observations whose managers continue to be single-tasking. We term this group as the non-switchers. In addition, we also create a control group for the acquired funds. In our sample, there are 202,893 fund-month observations that are not acquired by managers to multitask. We refer to these funds as the nonacquired funds. Note that the managers in the non-switcher group have to be single-tasking whereas the managers in the non-acquired funds can be single-tasking or multitasking. Therefore, the sample of non-acquired funds is much larger than the sample of the non-switchers.

B. Construction of Variables

We construct three variables to evaluate the risk-adjusted performance of the mutual funds in our sample. The first variable is the Sharpe ratio (*Sharpe*), calculated as a fund's annualized excess return over the risk-free rate divided by its annualized standard deviation (Sharpe (1966, 1994)). The second performance measure is the Carhart (1997) four-factor alpha (*Alpha*) estimated as the intercept of the following regression:

⁷ We exclude the cases (less than 1% of the sample) where managers take over more than four new-task funds as these are likely to be instances where a senior person's name (e.g., Bill Gross in case of PIMCO funds) is reported for administrative purposes.

$$R_{i,t} - R_{f,t} = \alpha_i + \beta_{i,m} \left(R_{m,t} - R_{f,t} \right) + \beta_{i,s} SMB_t + \beta_{i,h} HML_t + \beta_{i,MOM} MOM_t + \varepsilon_{i,t}$$
(1)

where $R_{i,t} - R_{f,t}$ is the return of the fund *i* in month *t* minus the risk free rate; $R_{m,t} - R_{f,t}$ is the excess return of the market over the risk free rate in month *t*; SMB_t is the return difference between small and large capitalization stocks in month *t*; HML_t is the return difference between high and low book-to-market stocks in month *t*, and MOM_t is the return difference between the stocks with high and low past returns in month *t*.⁸ The third performance measure is the Morningstar Category-adjusted alpha (*MS Alpha*) calculated as the intercept (α_t) from the following regression:

$$R_{i,j,t} - R_{f,t} = \alpha_i + \beta_{i,MS} \left(MS _ Index_{j,t} - R_{f,t} \right)$$
⁽²⁾

where $R_{i,j,t} - R_{f,t}$ is the return of fund *i*, following style *j*, in month *t* minus the risk-free rate and $MS _ Index_{j,t}$ is the average return of the funds following style *j* in month *t*.

Since the objective of the mutual fund companies is to maximize their assets, we quantify their economic incentives by estimating the net dollar flows, *i.e.*, the change in their total net assets over time, net of internal growth, under the assumption that all the dividends and other distributions are reinvested at the realized return:

$$EstimatedDollarFlows_{i,t} = TNA_{i,t} - TNA_{i,t-1}(1+R_{i,t})$$
(3)

where $TNA_{i,t}$ and $TNA_{i,t-1}$ are the total net assets of mutual fund *i* at time *t* and *t*-1, respectively and $R_{i,t}$ is the realized return earned by investors from time *t*-1 to *t*. We also compute an alternative measure, namely *N-SAR Flows*, using the *actual* net dollar flows reported

⁸ We thank Professor Kenneth French for making the returns on the market, risk-free rate, and the three factors (size, book-to-market, and momentum) available on his website: http://mba.tuck.dartmouth.edu/pages/faculty/ken.french/data_library.html.

by the mutual funds in their N-SAR forms filed with the Securities and Exchange Commission (SEC). Since the SEC started to require all the mutual funds to file N-SAR form in 1996, the measure *N-SAR Flows* is only available from January 1996 to December 2012.

II. Determinants of Managerial Multitasking

We begin our empirical investigation by analyzing the determinants of managers' switch to multitasking. To do so, we compare the performance of the funds whose managers switch from single-tasking to multitasking (*i.e.*, switchers) with the performance of the funds whose managers continue to manage a single fund (*i.e.*, non-switchers). Here and throughout the paper, we exclude the three months before and after the reported switch date to allow for the possibility that there is a transitional period where the incumbent manager continues to receive assistance running the acquired from the exiting manager. Results reported in panel A of Table I show that the funds managed by switchers outperform funds run by non-switchers over the 12-month period prior to the switch by 0.24, 1.63%, and 1.33% in terms of annual Sharpe ratio, alpha, and MS alpha, respectively. This finding indicates that managers who switch to multitasking exhibit superior performance prior to the switch.

We conduct a similar analysis for the acquired funds by comparing the performance of the acquired funds with that of the funds not acquired by managers to multitask (*i.e.*, non-acquired). Results in panel B of Table I show that the acquired funds underperform the non-acquired funds over the 12-month period prior to the switch by 0.10, 2.60%, and 2.46% using the *Sharpe*, *Alpha*, and *MS Alpha*, respectively. This finding suggests that one of the motives behind managerial multitasking is to turn around poorly performing funds by employing well-performing managers to take over these funds.

[Insert Table I Here]

In terms of other fund characteristics, we observe that the funds managed by the switchers have significantly greater turnover and attract greater investor flows compared to the funds managed by the non-switchers. We find that the acquired funds have significantly higher turnover and receive lower investor flows than the sample of non-acquired funds.

These univariate comparisons provide preliminary evidence that well-performing managers are more likely to switch from single-tasking to multitasking, and that the existing funds they acquire tend to be poorly performing. Next, we test whether this finding continues to hold in a multivariate setting after controlling for various fund characteristics. Such an analysis should also provide insights into the rationale for the mutual fund companies to initiate multitasking arrangements.

To examine the determinants in a multivariate framework, we estimate the following logistic regression modeling the type of incumbent funds from which the managers switch to multitasking:

$$Prob(y_{i,t}) = \beta'_{i} \overline{FundChar}_{i,[t-3,t-15]} + \alpha_{i} + \delta_{i} + \varepsilon_{i,t}$$

$$\tag{4}$$

where the dependent variable $y_{i,t}$ is an indicator variable equal to one if a manager *i* switches from single-tasking to multitasking in month *t* and zero if a manager continues to manage a single fund. *FundChar*_{*i*,[*t*-3,*t*-15]} is a vector of fund characteristics that includes a measure of the fund's risk-adjusted performance (*Sharpe*, *Alpha*, or *MS Alpha*) and its average total net assets, average expense ratio, average turnover ratio, and net dollar flows, all estimated or measured over the 12-month period prior to the switch after allowing for three-month gaps on either side of the switch in month *t*. We include both investment style (i.e., Morningstar Category) dummies, α_i , and time dummies, δ_i for style and time fixed effects. We also cluster the standard errors at the fund level in all the empirical tests.

[Insert Table II Here]

We report the results in models (1) - (3) of Table II. We find that managers who exhibit superior past performance are more likely to switch to multitasking. The estimated slope coefficients on all three performance measures are positive and statistically significant at the 1% level. In terms of the economic magnitude, a one-standard-deviation increase in past performance is associated with increases of 13.9 to 17.9% in the probability of a manager switching to multitasking. Regarding other fund characteristics, we find that the estimated slope coefficients on the fund's total net assets, the turnover ratio, and the net dollar flows are all positive and statistically significant. These additional characteristics indicate that the incumbent funds tend to be larger, exhibit more portfolio churning, and attract more investor flows compared to the funds managed by the non-switchers. These findings corroborate our univariate results in Table I.

Having examined the characteristics of the incumbent funds, we proceed to investigate the characteristics of the acquired funds. Khorana (1996) documents an inverse relation between the probability of managerial replacement and fund's past performance. Motivated by his finding, we hypothesize that funds are more likely to be acquired by managers to multitask if they perform poorly. Models (4) – (6) of Table II report the results of the logistic regressions modeling the type of existing funds that are acquired by managers to multitask. The dependent variable is an indicator variable that equals one if a fund is acquired by managers to multitask in

month *t* and zero otherwise. The independent variables are identical to those used in analyzing the determinants of the incumbent funds in models (1) - (3) of Table II.

Consistent with our hypothesis, we find that funds are more likely to be acquired by managers to multitask subsequent to poor performance. The estimated slope coefficients on all three performance measures are negative and highly significant. In terms of the economic magnitude, a one-standard-deviation increase in fund performance reduces the probability of the fund being acquired by 11.3% to 25.8%. Further, we find positive estimated slope coefficients on the fund's total net assets and the expense ratio. The coefficient on the net dollar flows, however, is negative. These results suggest that the acquired funds tend to be larger, have higher turnover, and experience lower investor flows compared to non-acquired funds. Again, these findings are consistent with the univariate results in Table I.

Overall, the results from both the univariate and multivariate analyses in this section show that managers who switch from single-tasking to multitasking exhibit superior past performance in the incumbent funds prior to the switch. Moreover, the existing funds they acquire to multitask tend to be poorly performing. We offer three explanations for these findings. First, well-performing managers of incumbent funds can create a positive spillover effect in form of greater investor flows into the acquired funds. Similar spillover effects have been documented in the context of star funds in fund families (Nanda, Wang, and Zheng, 2004) and reputable managers launching new funds (Chen and Lai, 2010). Second, enlisting well-performing managers to multitask can help fund companies to turn around their poorly performing funds, whose presence can adversely affect companies' reputation. There can be other benefits of replacing the managers of poorly performing funds. For example, Lynch and Musto (2003) theoretically model and empirically test the decrease in the flow-performance sensitivity subsequent to manager turnover. They show that investors are less likely to withdraw from poorly performing funds if the managers are replaced. Finally, since multitasking increases the manager's span of control, mutual fund companies can use it to retain their good managers and to replace their bad managers, thereby maximizing the economic surplus through their monitoring role (Gervais, Lynch, and Musto, 2005).

III. Managerial Multitasking and Fund Performance

In this section, we examine the implications of managerial multitasking for fund performance, for which we have three hypotheses. Our first hypothesis, *distraction hypothesis*, predicts that both the performance of the incumbent and the acquired funds suffer due to managerial multitasking since managers running multiple funds simultaneously are more likely to be distracted and overstretched. Our second hypothesis, which we refer to as the *effort diversion hypothesis*, is that multitasking managers can divert their attention and effort from the existing funds to the new task, resulting in performance deterioration for the incumbent funds but improvement of performance in the acquired funds. Our third and final hypothesis, *synergy creation hypothesis*, predicts that the performance of both the incumbent and acquired funds improve after the managers' switch to multitasking. The underlying premise is that by managing multiple funds simultaneously, multitasking managers can exploit the synergistic benefits associated with learning while conducting investment research for multiple funds.

[Insert Table III Here]

To test these three hypotheses, we start by examining the changes in both incumbent and acquired funds' performance for the two-year period around the managers' switch to multitasking.⁹ The univariate results reported in Table III suggest that there is a striking decline in the risk-adjusted performance of the incumbent funds over the two-year period (one year on either side of managers' switch to multitasking). All three performance measures are significantly lower in the period after the managers begin multitasking. Annual *Sharpe*, *Alpha*, and *MS Alpha* decline by 0.206, 1.82%, and 1.43% respectively in the post-switch one-year period. In contrast to the incumbent funds, there is a statistically significant improvement in the performance of the acquired funds over the same one-year period after the switch: 0.206, 2.28%, and 2.97% increases in the annual *Sharpe*, *Alpha*, and *MS Alpha*, respectively.

To corroborate these univariate results, we next estimate the following multivariate regression modeling the risk-adjusted performance over the two-year period around managers' switch to multitasking:

$$Perf_{i,t} = \lambda_i After_{i,t} + \phi_i FundChar_{i,[t-1,t-12]} + \alpha_i + \delta_t + \xi_{i,t}$$
(5)

To estimate the above regression, we construct a panel of fund-year observations from both before and after the switch. The dependent variable in the regression is the risk-adjusted performance for fund *i* at time *t*, $Perf_{i,t}$, where *t* is either the one-year period before or after the switch. The main independent variable of interest is an indicator variable, $After_{i,t}$, which equals one if the fund-year observation is from the one-year period after the switch and zero if the fundyear observation is from the one-year period before the switch. The estimated slope coefficient λ_i on $After_{i,t}$ captures the impact of the switch on fund performance. We include a vector of average fund characteristics $\overline{FundChar_{i,[t-1,t-12]}}$ such as fund's total net assets, expense ratio,

⁹ Throughout the paper, we focus on the two-year period around the managers' switch to multitasking. Analysis over a longer period will impose significant survivorship basis, in addition to substantially reducing the sample because the mean and median manager tenure in our sample is 3.7 and 5.1 years, respectively.

turnover ratio, and net dollar flows. Finally, we also include both style and time fixed effects, α_i and δ_i .

[Insert Table IV Here]

We report our findings in Table IV. Consistent with the univariate results in Table III, we find that the performance of the incumbent funds deteriorates while the performance of acquired funds improves after the switch. The estimated slope coefficients on *After* are negative and highly significant for the incumbent funds using the three performance measures (-0.126, -1.766, and -1.227) while the coefficients are significantly positive for the acquired funds (0.209, 1.914, and 2.889). These coefficients provide the economic impact of the multitasking arrangement on fund performance. For example, the annual *Alpha* of the incumbent (acquired) funds declines (improves) by 1.766% (1.914%) as a result of managerial multitasking. These performance changes are also economically significant considering the average annual *Alphas* of the incumbent and acquired funds prior to the switch are 1.39% and -2.15%, respectively (see Table I).

Taken together, both the univariate and multivariate analyses in Tables III and IV show a decline in the incumbent funds' performance and an improvement in the acquired funds' performance. We interpret these results as being consistent with the effort diversion hypothesis, and not in favor of either the distraction hypothesis or the synergy creation hypothesis.

A. Matched Sample Analysis

There are two potential concerns with our findings in Tables III and IV. First, the performance deterioration of previously well-performing incumbent funds and the performance

improvement of previously poorly performing acquired funds can simply be due to mean reversion in fund performance. In other words, the observed change in fund performance would have happened even if the manager did not switch to multitasking. Second, since we observe that the incumbent funds tend to be larger and have received greater investor flows, the decline in their performance after the switch can be potentially driven by decreasing returns to scale documented in Berk and Green (2004), Chen et al. (2004), and Pástor and Stambaugh (2012) and may have little to do with the diversion of managerial effort.

[Insert Table V Here]

To address these two concerns, we conduct matched-sample analyses by investigating the change in the risk-adjusted performance of the funds that share similar characteristics with the incumbent funds and the acquired funds except that their managers are not involved in multitasking. In particular, we construct control samples by matching the incumbent and acquired funds using the propensity score estimates from the results of the logistic regressions modeling the switch (see Table II, Models 1 and 4). We then estimate the same multivariate regressions used in Table IV using the matched control samples and compare the coefficients on *After* for the multitasking funds and the control funds. For robustness, we also use random matching to construct a second set of control funds.

We report our findings in Table V. Panel A contains the results for the incumbent funds and Panel B contains the results for the acquired funds. Two findings merit mention. First, the coefficients on *After* for the control samples of funds are uniformly insignificant at conventional levels, regardless of whether we construct our control samples using propensity scores or through random matching. Second, we use *F*-tests to compare the differences of the change in performance of the treated group to that of the control group and find that, in general, the differences are statistically significant. For example, using the results of the tests that compare the performance of the treatment funds to that of the propensity-score-matched samples of funds, incumbent (acquired) funds experience a 1.93% (2.59%) larger drop (increase) in *Alpha*. These results help allay concerns that our earlier findings are not driven by either mean reversion or decreasing returns to scale.

B. Further evidence supporting the effort diversion hypothesis

Our findings so far show that the performance of the incumbent funds deteriorates after their managers begin to multitask while the performance of the acquired funds improves. We now turn our attention to uncovering further evidence supporting the effort diversion hypothesis. If the incumbent funds' performance suffers because managers are diverting effort away from the incumbent funds to their newly acquired funds, we should expect that the magnitude of performance deterioration to be increasing in the level of effort diverted. However, since the manager's effort level is not directly observable, we use three proxies for the level of effort directed towards the acquired fund. The first variable we use is *New Fund*, an indicator variable equal to one if the acquired fund is a newly-launched fund and zero if the acquired fund is preexistent. Prior literature finds that flow-performance sensitivity is higher for younger funds than it is for older funds (e.g., Chevalier and Ellison, 1997). Given this evidence, multitasking managers acquiring new funds may allocate more effort to the acquired fund, relative to managers acquiring existing funds, to ensure that these new funds perform well and attract flows.

Our second proxy for effort is *Acquired Fund Turnover*. Portfolio turnover measures how frequently the securities in a portfolio are bought or sold and thus captures, to some extent, the

amount of effort a manager expends managing the acquired fund.¹⁰ Finally, our third and final measure of managerial effort is *Acquired Style Deviation*, which equals the absolute value of the acquired fund's beta with respect to its Morningstar style index in a one-factor model (e.g., its $\beta_{i,MS}$ from Equation 2 above) minus 1. This measure is similar in spirit to the Active Share measure of Cremers and Petajisto (2009), which is designed to measure how much a fund deviates from its benchmark. We argue that the level of effort a manager expends on the acquired fund is increasing in the degree of the acquired fund's return deviation from its benchmark.

To test the prediction that the magnitude of the incumbent funds' performance drop is increasing in the level of effort the managers allocate to their acquired funds, we estimate the following multivariate regression:

$$Perf_{i,t} = v_i \left(After \times Effort Proxy_{i,t} \right) + \kappa_i After_{i,t} + \omega_i \overline{FundChar}_{i,[t-1,t-12]} + \alpha_i + \delta_t + \xi_{i,t}$$
(6)

As in the case of equation (5), we include observations both before and after the managers' switch to multitasking. The dependent variable is incumbent fund *i*'s one-year risk-adjusted performance, $Perf_{i,t}$, at time *t*. The independent variable of interest is *After* × *EffortProxy*, which is the interaction of *After* and one of our three proxies for effort. The estimated slope coefficient, v_i , on *After* × *EffortProxy* captures the incremental impact of greater effort diversion on the incumbent fund's performance. We expect the coefficient on *After* × *EffortProxy* to be negative if greater effort diversion is associated with more performance deterioration in the incumbent

¹⁰ We acknowledge that higher turnover can also be associated with undesirable practices such as window dressing (Musto, 1997, 1999; Agarwal, Gay, and Ling, 2014) and therefore turnover may not necessarily capture effort. That said, many studies find turnover and other measures of a manager's activeness to be positively related to fund performance (Wermers, 2000; Cremers and Petajisto, 2009; Amihud and Goyenko, 2013). To the extent that managerial effort is positively correlated with fund performance, manager's effort should be at least somewhat reflected in fund turnover.

funds. The vector *FundChar* includes all of the variables used earlier in equation (5) (e.g., average fund size, expense ratio, turnover, and flows) and α_i and δ_i are style and year fixed effects, respectively.

[Insert Table VI Here]

We report the results in Table VI. Panels A, B, and C contain the results with *New Fund Acquired Fund Turnover*, and *Acquired Style* Deviation as proxy for effort, respectively. The coefficient on *After* × *EffortProxy* is negative in all specifications and statistically significant at the 5% level in seven out of nine cases. These findings suggest that the decline in the riskadjusted performance of incumbent funds is greater when managers take over funds in which they devote greater effort to manage. This evidence provides further support to the effort diversion hypothesis.

IV. Managerial Multitasking and Fund Flows

In this section, we examine the economic incentives of the mutual fund companies to engage in multitasking by analyzing its effect on the investor flows. In the previous section, we have shown that when portfolio managers switch from single-tasking to multitasking, they divert their effort from the incumbent funds to the acquired funds. As a result, the incumbent funds experience significant performance deterioration over a 12-month window following the switch, while the performance of the acquired funds improves. If investors of the incumbent funds can anticipate the adverse effects of multitasking on future performance, investor flows should decrease for these funds. In contrast, we posit that investor flows into the acquired funds should increase due to the positive spillover effect of well-performing multitasking managers. For multitasking to be a profitable arrangement, the marginal benefits of engaging in multitasking should exceed the marginal costs for fund companies. Therefore, we predict that the net impact on dollar flows into the fund companies should be positive.

We test these three predictions by estimating multivariate regressions modeling investor flows before and after managers' switch to multitasking for both the incumbent and acquired funds. The specification is similar to the one used in the previous section for examining the changes in fund performance around the switch to multitasking in equation (5). The dependent variable is the estimated dollar flows as defined in equation (3). The main independent variable of interest is *After*, an indicator variable which equals one if the observation is from the 12month period after the switch and zero if the observation is from the 12-month period before the switch. The coefficient on *After* therefore captures the impact of the switch on the investor flows. We control for various fund characteristics that have been previously shown to affect fund flows. These characteristics include the contemporaneous and lagged risk-adjusted performance, the fund's total net assets, the expense ratio, and the turnover ratio. Note that in our empirical tests, we control for both contemporaneous and past performance, which implies that any effect on fund flows stems from investors' response to how multitasking affects *future* fund performance. This intuition underlies our hypotheses outlined above.

We report the results in models (1) and (3) of Table VII. Contrary to our prediction, we find no significant change in the estimated dollar flows of incumbent funds after the switch as the coefficient on *After* is positive but insignificant (coeff. = -2.956, *t*-stat = -0.21). It is conceivable that investors may not be able to fully anticipate the undesirable consequences of multitasking on the future performance of incumbent funds. Another potential explanation for this unexpected result is that the investors of the incumbent funds may regard multitasking as a

signal of the manager's quality and/or importance in the fund companies and thus remain invested in the manager's incumbent fund.

In contrast to the incumbent funds and consistent with our prediction, the acquired funds experience a significant increase in the investor flows after the managers' switch to multitasking. The coefficient on *After* is 22.397, significant at the 5% level, suggesting an increase of about \$22.40 million net dollar flows for the acquired funds. This increase in the investor flows is economically significant as the acquired funds experience a *negative* 30.95 million net dollar flows in the 12-month period before being acquired (see panel B of Table I). This finding is consistent with the idea that these funds experience a positive spillover effect when they are taken over by well-performing managers who begin multitasking.

[Insert Table VII Here]

Our results so far are based on *estimated* dollar flows using equation (3). For robustness, we employ an alternative dollar flow measure, *N-SAR Flows*, using the *actual* monthly flows reported by mutual funds to the SEC in the N-SAR forms since 1996. Despite the drop in the observations, in results reported in models (2) and (4) in Table VII, we continue to find a positive but insignificant coefficient (coeff. = 15.256, *t*-stat = 0.58) on the *After* variable for the incumbent funds and a positive and significant coefficient (coeff. = 30.364, *t*-stat = 2.04) for the acquired funds.

Next, we test whether the positive spillover effect of multitasking managers also impacts the new funds launched by these managers. In particular, we compare the net dollar flows into the new funds managed by multitasking managers versus those managed by single-tasking managers. Note that since there is no prior data on new funds, it is not possible to do a timeseries analysis of changes in investor flows as we do for the existing funds acquired by the multitasking managers. Instead, we estimate the following cross-sectional regression:

$$\$Flows_i = \theta_i Multitasking_i + \rho_i FundChar_i + \alpha_i + \delta_i + \varsigma_i$$
(6)

where the dependent variable $Dollarflow_i$ is either the *Estimated* or the *N-SAR Flows* over the 12-month window after the launch of a new fund. The main independent variable of interest is an indicator variable, *Multitasking*, that equals one if a new fund is launched by a manager to multitask and zero if the new fund is the only fund managed by a manager. We again control for the same set of fund characteristics as in our previous regressions with the exception of past performance and size since these variables are not available for the new funds.¹¹

[Insert Table VIII Here]

The results reported in Table VIII confirm a positive spillover effect of multitasking managers on the new funds launched by them. On average, new funds launched by multitasking managers attract \$9.27 million and \$14.53 million greater estimated and actual dollar flows, respectively, compared to the funds launched by single-tasking managers. Taken together, the asymmetry in the responses of the investors of the incumbent funds and the new-task funds (*i.e.*, acquired funds and new funds) allows the fund companies to increase their assets.¹²

V. Switchback from Multitasking to Single-tasking

As discussed in the previous section, mutual fund appear to trade off the benefit of higher acquired fund flows with the cost of lower incumbent fund performance when deciding to initiate

¹¹ Since all the new funds have zero assets under management at inception, the fund size and fund flows are the same.

¹² We also estimate these regressions using the piecewise specification of Sirri and Tufano (1998). The results are similar to those presented in Table VIII and are available upon request.

multitasking arrangements. In well-functioning financial markets with rational investors, incumbent funds' investors should recognize the negative externalities associated with multitasking and withdraw their capital from underperforming funds at some point in the future. These outflows should, in turn, induce the fund companies to terminate these multitasking arrangements. To test this hypothesis, we examine the determinants of 351 cases where managers *switchback* from multitasking to single-tasking. We estimate logistic regressions similar to those reported in Table 2 using the dependent variable, *Switchback*, an indicator variable that equals one in the month a manager stops multitasking and zero otherwise. The independent variables include the incumbent fund's performance, flows, turnover, expense ratio, and size.

The results are reported in Panel A of Table IX. Consistent with our intuition, fund companies are more likely to terminate multitasking arrangements when the incumbent funds' performance is especially low and when funds experience outflows. The coefficients on all three risk-adjusted performance measures as well as past flows are negative and statistically significant. For instance, a one-standard-deviation decrease in the incumbent fund's annual *Sharpe (Flows)* increases the probability of a multitasking arrangement being terminated by 16.09% (20.95%).

[Insert Table IX here]

We now examine the impact of the termination of the multitasking arrangements on the performance of the funds the managers retain. The findings in Section III consistently suggest that, when managers switch from single-tasking to multitasking, they divert their effort away from the incumbent funds which leads to performance deterioration for these funds. To further test the effort diversion hypothesis, we examine if the converse is true, *i.e.*, when managers switchback, is there an improvement in the performance of the funds retained by these managers subsequent to the switchback? We conduct similar multivariate analysis as in Table IV for the fund performance after the switchback, and report our findings in Panel B of Table IX. We find the coefficient on *After* is positive and significant for all three performance measures: 0.154, 1.666, and 1.159 for *Sharpe*, *Alpha* and *MS Alpha* respectively. The fact that the risk-adjusted performance of the funds retained by the managers who switchback improves significantly provides further support for the effort diversion hypothesis.

Taken together, the results in this section show that agency problems associated with multitasking do not persist as investors of incumbent funds rationally respond by withdrawing their capital which forces fund companies to terminate the multitasking arrangements. Moreover, we observe an improvement in the performance of the funds retained by the managers after discontinuance of multitasking which further corroborates the effort diversion hypothesis.

VI. Concluding Remarks

In this paper, we investigate the determinants and consequences of managerial multitasking in the mutual fund industry. Our empirical analyses reveal four notable findings. First, fund companies select well-performing managers to multitask to either turn around poorly performing funds or to launch new funds. Second, when managers multitask, the performance of the incumbent funds declines while that of the acquired funds improves during the 12-month period subsequent to multitasking. Third, while incumbent funds experience no changes in the investor flows, the acquired funds and new funds attract more flows subsequent to multitasking. As a result, mutual fund companies benefit in terms of greater aggregate investor flows and more

assets under management. This advantage is in addition to the other benefits fund companies receive from multitasking such as the improvement of their struggling funds, the retention of their superior managers, and the launching of new funds. These benefits, however, come at the expense of the investors of the incumbent funds. Finally, we observe that investors of incumbent funds respond rationally by withdrawing their capital, which leads to the termination of these multitasking arrangements.

Taken together, these findings suggest potential agency problems that arise when mutual fund managers multitask. The fact that some investors are adversely affected, at least in the short run, by their manager's distorted incentives has policy implications for the regulatory bodies governing the mutual fund industry. For instance, investors may benefit from more transparent and timelier disclosure about the changes in their fund managers' workload. Our study also sheds light on the pivotal role played by fund companies in determining the span of control for their portfolio managers and the internal allocation of their managerial resources, including the replacement of poorly performing managers and the retention of well-performing managers.

REFERENCES

Agarwal, Vikas, Gerald D. Gay, and Leng Ling, 2014, Window dressing in mutual funds, *Review of Financial Studies* 27, 3133–3170.

Agarwal, Vikas, Yan Lu, and Sugata Ray, 2015, Under one roof: A study of simultaneously managed hedge funds and funds of hedge funds, *Management Science* forthcoming.

Amihud, Yakov, and Ruslan Goyenko, 2013, Mutual fund's R^2 as predictor of performance, *Review of Financial Studies* 26, 667–694.

Ashraf, Rasha, Narayanan Jayaraman, and Harley E. Ryan Jr., 2012, Do pension-related business ties influence mutual fund proxy voting? Evidence from shareholder proposals on executive compensation, *Journal of Financial and Quantitative Analysis* 47, 567–588.

Baer, Michaela, Alexander Kempf, and Stefan Ruenzi, 2011, Is a team different from the sum of its parts? Evidence from mutual fund managers, *Review of Finance* 15, 359–396.

Berk, Jonathan B., and Richard C. Green, 2004, Mutual fund flows and performance in rational markets, *Journal of Political Economy* 112, 1269–1295.

Bhattacharya, Utpal, Jung Hoon Lee, and Veronika Krepely Pool, 2013, Conflicting family values in mutual fund families, *Journal of Finance* 68, 173–200.

Bliss, Richard T., Mark E. Potter, and Christopher Schwarz, 2008, Performance characteristics of individual vs. team managed mutual funds, *Journal of Portfolio Management* 34, 110–119.

Brown, Keith C., W. Van Harlow, and Laura T. Starks, 1996, Of tournaments and temptations: An analysis of managerial incentives in the mutual fund industry, *Journal of Finance* 51, 85–110.

Carhart, Mark M., 1997, On persistence in mutual fund performance, *Journal of Finance* 52, 57–82.

Chen, Joseph, Harrison Hong, Ming Huang, and Jeffrey D. Kubik, 2004, Does fund size erode mutual fund performance? The role of liquidity and organization, *American Economic Review* 94, 1276–1302.

Chen, Joseph, Harrison Hong, and Jeffrey D. Kubik, 2013, Outsourcing mutual fund management firm boundaries, incentives and performance, *Journal of Finance* 68, 523–558.

Chen, Li-Wen, and Fan Chen, 2009, Does concurrent management of mutual and hedge funds create conflicts of interest? *Journal of Banking and Finance* 33, 1423–1433.

Chen, Hsuan-Chi, and Christine W. Lai, 2010, Reputation stretching in mutual fund starts, *Journal of Banking and Finance* 34, 193–207.

Chevalier, Judith, and Glenn Ellison, 1997, Risk taking by mutual funds as a response to incentives, *Journal of Political Economy* 105, 1167–1200.

Choi, Darwin, Bige Kahraman, and Abhiroop Mukherjee, 2014, Learning about mutual fund managers, Working paper, Hong Kong University of Science & Technology and Stockholm School of Economics.

Cici, Gjergji, Scott Gibson, and Rabih Moussawi, 2010, Mutual fund performance when parent firms simultaneously manage hedge funds, *Journal of Financial Intermediation* 19, 169–187.

Cohen, Lauren, and Breno Schmidt, 2009, Attracting flows by attracting big clients, *Journal of Finance* 64, 1225–1252.

Cremers, K. J. M., and Antti Petajisto, 2009, How active is your fund manager? A new measure that predicts performance, *Review of Financial Studies* 22, 3329–3365.

Davis, Gerald F., and E. Han Kim, 2007, Business ties and proxy voting by mutual funds, *Journal of Financial Economics* 85, 552–570.

Deuskar, Prachi, Joshua M. Pollet, Z. Jay Wang, and Lu Zheng, 2011, The good or the bad? Which mutual fund managers join hedge funds? *Review of Financial Studies* 24, 3008–3024.

Ferris, Stephen P., Murali Jagannathan, and A. C. Pritchard, 2003, Too busy to mind the business? Monitoring by directors with multiple board appointments, *Journal of Finance* 58, 1087–1111.

Fich, Eliezer M., and Anil Shivdasani, 2006, Are busy boards effective monitors? *Journal of Finance* 61, 689–724.

Gaspar, Jose-Miguel, Massimo Massa, and Pedro Matos, 2006, Favoritism in mutual fund families? Evidence on strategic cross-fund subsidization, *Journal of Finance* 61, 73–104.

Gervais, Simon, Anthony W. Lynch, and David K. Musto, 2005, Fund families as delegated monitors of money managers, *Review of Financial Studies* 18, 1139–1169.

He, Jia, Lilian Ng, and Qinghai Wang, 2004, Quarterly trading patterns of financial institutions, *Journal of Business* 77, 493–509.

Hu, Ping, Jayant R. Kale, Marco Pagani, and Ajay Subramanian, 2011, Fund flows, performance, managerial career concerns, and risk-taking, *Management Science* 57, 628–646.

Huang, Jennifer, Clemens Sialm, and Hanjiang Zhang, 2011, Risk shifting and mutual fund performance, *Review of Financial Studies* 24, 2575–2616.

Jayaraman, Narayanan, Ajay Khorana, and Edward Nelling, 2002, An analysis of the determinants and shareholder wealth effects of mutual fund mergers, *Journal of Finance* 57, 1521–1551.

Kempf, Alexander, and Stefan Ruenzi, 2008, Tournaments in mutual fund families, *Review of Financial Studies* 21, 1013–1036.

Kempf, Alexander, Stefan Ruenzi, and Tanja Thiele, 2009, Employment risk, compensation incentives, and managerial risk taking: Evidence from the mutual fund industry, *Journal of Financial Economics* 92, 92–108.

Khorana, Ajay, 1996, Top management turnover: an empirical investigation of mutual fund managers, *Journal of Financial Economics* 40, 403–427.

Lakonishok, Josef, Andrei Shleifer, Richard Thaler, and Robert Vishny, 1991, Window dressing by pension fund managers, *American Economic Review* 81, 227–231.

Lynch, Anthony W., and David K. Musto, 2003, How investors interpret past fund returns, *Journal of Finance* 58, 2033–2058.

Massa, Massimo, Jonathan Reuter, and Eric Zitzewitz, 2010, When should firms share credit with employees? Evidence from anonymously managed mutual funds, *Journal of Financial Economics* 95, 400–424.

Meier, Iwan, and Ernst Schaumburg, 2006, Do funds window dress? Evidence for U.S. domestic equity mutual funds, Working paper, HEC Montreal and Kellogg School of Management.

Musto, David K., 1997, Portfolio disclosure and year-end price shift, *Journal of Finance* 52, 1563–1588.

Musto, David K., 1999, Investment decisions depend on portfolio disclosures, *Journal of Finance* 54, 935–952.

Nanda, Vikram, Z. Jay Wang and Lu Zheng, 2004, Family values and the star phenomenon: strategies of mutual fund families, *Review of Financial Studies* 17, 667–698.

Ng, Lilian, and Qinghai Wang, 2004, Institutional trading and the turn-of-the-year effect, *Journal of Financial Economics* 74, 343–366.

Nohel, Tom, Z. Jay Wang, and Lu Zheng, 2010, Side-by-side management of hedge funds and mutual funds, *Review of Financial Studies* 23, 2342–2373.

Pástor, Luboš, and Robert F. Stambaugh, 2012, On the size of the active management industry, *Journal of Political Economy* 12, 740–781.

Patel, Saurin, and Sergei Sarkissian, 2014, To group or not to group? Evidence from mutual funds, Working paper, McGill University.

Sandhya, Vallapuzha, 2012, Agency problems in target-date funds, Working paper, Georgia State University.

Schwarz, Christopher G., 2012, Mutual fund tournaments: The sorting bias and new evidence, *Review of Financial Studies* 25, 913–936.

Sharpe, William F., 1966, Mutual fund performance, Journal of Business 39, 119–138.

Sharpe, William F., 1994, The Sharpe ratio, Journal of Portfolio Management 21, 49-58.

Sirri, Erik R. and Peter Tufano, 1998, Costly search and mutual fund flows, *Journal of Finance* 53, 1589-1622.

Sun, Zheng, Ashley Wang, and Lu Zheng, 2012, The road less traveled: Strategy distinctiveness and hedge fund performance, *Review of Financial Studies* 25, 96–143.

Wermers, Russ, 2000, Mutual fund performance: An empirical decomposition into stock-picking talent, style, transactions costs, and expenses, *Journal of Finance* 55, 1655–1695.

Yadav, Vijay, 2011, Portfolio matching by multi-fund managers: Effects on fund performance and flows, Working paper, ESSEC Singapore.

Table I: Characteristics of Incumbent and Acquired Funds

Panel A compares the characteristics of the funds whose managers switch from single-tasking to multitasking (*i.e.*, switchers) with those of the funds whose managers continue to manage a single fund (*i.e.*, non-switchers). The differences between the characteristics of the switchers and non-switchers are reported in the last column. Panel B compares the characteristics of the acquired funds (*i.e.*, acquired) with those of the funds that are not acquired by managers to multitask (*i.e.*, non-acquired). The differences between the characteristics of the acquired and the non-acquired funds are reported in the last column. Reported fund characteristics include the three measures of risk-adjusted performance (Sharpe ratio, Carhart (1997) four-factor alpha (in % p.a.), and Morningstar Category-adjusted alpha (in % p.a)), the fund's average total net assets (in millions of dollars), the average expense ratio (in %), the average turnover ratio (in %), and the net dollar flows (in millions of dollars), all estimated or measured over a 12-month window prior to the month of the switch. All variables are winsorized at the 1st and the 99th percentile levels. Our sample period is from January 1980 to December 2012. Statistical significance of 1%, 5%, and 10% is indicated by ***,**, and * respectively.

Fund Characteristic	Switchers	Non-switchers	Difference
Sharpe Ratio	0.835	0.596	0.239***
Four–Factor Alpha	1.39%	-0.24%	1.63%***
Morningstar Alpha	1.68%	0.35%	1.33%***
Net Assets (\$ Millions)	699.76	630.60	69.16
Expense Ratio (%)	1.37	1.37	0.00
Turnover (%)	99.00	91.00	8.00**
Net Flows (\$ Millions)	57.86	19.38	38.48***

Panel A: Incumbent Funds

Panel B: Acquired Funds

Fund Characteristic	Acquired	Non-Acquired	Difference
Sharpe Ratio	0.464	0.563	-0.099
Four–Factor Alpha	-2.15%	0.45%	-2.60%***
Morningstar Alpha	-1.93%	0.53%	-2.46%***
Net Assets (\$ Millions)	668.35	669.93	-1.58
Expense Ratio (%)	1.42	1.37	0.05
Turnover (%)	1.20	0.95	0.25***
Net Flows (\$ Millions)	-30.95	27.22	-58.17***

Table II: Determinants of Incumbent and Acquired Funds

This table reports the results of logistic regressions modeling the type of incumbent funds from which the managers switch from single-tasking to multitasking (models (1) - (3)) and the type of existing funds acquired by those managers to multitask (models (4) - (6)) over the sample period of January 1980 to December 2012. In models (1) - (3), the dependent variable is an indicator variable that equals one if a manager switches from single-tasking to multitasking in month *t* and zero if a manager continues managing a single fund. In models (4) - (6), the dependent variable is an indicator variable that equals one if a fund is acquired by a manager to multitask in month *t* and zero otherwise. The independent variables include one of the three measures of annual risk-adjusted performance (Sharpe ratio, Carhart (1997) four-factor alpha (in % p.a.), or Morningstar Category-adjusted alpha (in % p.a.)), the natural logarithm of the fund's average total net assets (in millions of dollars), the average expense ratio (in %), the average over the 12-month period three months prior to the month of the switch. All variables are winsorized at the 1st and the 99th percentile levels. We control for investment style and year fixed effects. The standard errors are clustered by fund. The *t*-statistics are reported in the parentheses. Statistical significance of 1%, 5%, and 10% is indicated by ***,**, and * respectively.

	Incumbent Funds			A	Acquired Funds			
	(1)	(2)	(3)	(4)	(5)	(6)		
Sharpe Ratio	0.135***			-0.178***				
	(3.02)			(-2.62)				
Four-Factor Alpha		0.013***			-0.013**			
		(3.06)			(-2.00)			
Morningstar Alpha			0.021***			-0.042***		
			(3.77)			(-4.25)		
Ln Assets (\$millions)	0.114***	0.116***	0.116***	0.070*	0.069*	0.072*		
	(4.21)	(4.30)	(4.27)	(1.76)	(1.73)	(1.81)		
Avg. Expense Ratio	0.123	0.118	0.128	0.047	0.054	0.014		
	(1.08)	(1.04)	(1.13)	(0.34)	(0.39)	(0.10)		
Avg. Turnover	0.001**	0.001**	0.001**	0.001**	0.001**	0.001*		
	(2.10)	(2.10)	(2.01)	(2.10)	(2.03)	(1.77)		
Ln Flows ((\$millions)	0.010***	0.010***	0.009***	-0.016***	-0.017***	-0.014***		
	(3.63)	(3.74)	(3.45)	(-4.11)	(-4.29)	(-3.79)		
Style FE	YES	YES	YES	YES	YES	YES		
Year FE	YES	YES	YES	YES	YES	YES		
Observations	83,299	83,299	83,299	203,191	203,191	203,191		
Pseudo R ²	0.0181	0.0181	0.0182	0.0391	0.0385	0.0433		

Table III: Change in Fund Characteristics Before and After the Switch

Panel A (Panel B) reports the characteristics of the incumbent (acquired) funds prior to the switch (*i.e.*, month *t*-15 to *t*-3) and after the switch (*i.e.*, month *t*+3 to *t*+15). The change in the fund characteristics from the pre-switch period (*i.e.*, month *t*-15 to *t*-3) to the post-switch period (*i.e.*, month *t*+3 to *t*+15) are reported in the last column. Reported fund characteristics include three measures of risk-adjusted performance (Sharpe ratio, Carhart (1997) four-factor alpha (in % p.a.), and Morningstar Category-adjusted alpha (in % p.a)), the fund's average total net assets (in millions of dollars), the average expense ratio (in %), the average turnover ratio (in %), and the net dollar flows (in millions of dollars), all estimated or measured over 12 months prior to the switch (*i.e.*, month *t*-15 to *t*-3) and 12 months after the switch (*i.e.*, month *t*+3 to *t*+15) after leaving three months around the switch in month *t*. All variables are winsorized at the 1st and the 99th percentile levels. Our sample period is January 1980 to December 2012. Statistical significance of 1%, 5%, and 10% is indicated by ***,**, and * respectively.

Panel A: Incumbent Funa	Panel A	: Incun	ıbent I	Funds
-------------------------	---------	---------	---------	-------

Fund Characteristic	Before	After	Difference
Sharpe Ratio	0.835	0.629	-0.206***
Four–Factor Alpha	1.39%	-0.73%	-1.82%***
Morningstar Alpha	1.69%	0.26%	-1.43%***
Net Assets (\$ Millions)	699.76	910.59	210.83**
Expense Ratio (%)	1.37	1.37	0.00
Turnover (%)	0.993	0.972	-2.05
Net Flows	55.76	56.77	1.01

Panel B: Acquired Funds

Fund Characteristic	Before	After	Difference
Sharpe Ratio	0.464	0.670	0.206*
Four–Factor Alpha	-2.03%	0.25%	2.28%**
Morningstar Alpha	-2.09%	0.88%	2.97%***
Net Assets (\$ Millions)	668.35	667.84	-0.51
Expense Ratio (%)	1.42%	1.40%	-0.02%
Turnover (%)	120.30%	1.29	8.94%
Net Flows	-28.79	-17.41	11.38

Table IV: Multivariate Analysis of the Changes in Fund Performance after the Switch

This table reports the changes in the risk-adjusted performance of the incumbent funds (models (1) - (3)) and the acquired funds (models (4) - (6)) before (*i.e.*, month *t*-15 to *t*-3) and after (*i.e.*, month *t*+3 to *t*+15) the switch over the sample period from January 1980 to December 2012. The dependent variable in models (1) and (4) is the one-year Sharpe ratio calculated over the 12-month window after the switch. The dependent variable in models (2) and (5) is the one-year Carhart (1997) four-factor alpha estimated over the 12-month window after the switch. The dependent variable in models (3) and (6) is the one-year Morningstar Category-adjusted alpha estimated over the 12-month window after the switch. The main independent variable of interest is *After*, an indicator variable that equals one (zero) if the observation is within the 12-month period after (before) the managers' switch to multitasking. Other independent variables include the natural logarithm of the fund's average total net assets (in millions of dollars), the average expense ratio (in %), the average turnover ratio (in %), and the natural logarithm of net dollar flows (in millions of dollars), all estimated or measured over the 12-month windows before and after the switch. All variables are winsorized at the 1st and the 99th percentile levels. We control for investment style and year fixed effects. The standard errors are clustered at the fund level. The *t*-statistics are reported in the parentheses. Statistical significance of 1%, 5%, and 10% is indicated by ***,**, and * respectively.

	Incumbent Funds			Acquired Funds			
	(1)	(2)	(3)	 (4)	(5)	(6)	
	Sharpe	Alpha	MS Alpha	 Sharpe	Alpha	MS Alpha	
After	-0.126**	-1.766***	-1.227***	0.209**	1.914**	2.889***	
	(-2.14)	(-2.96)	(-2.91)	(2.56)	(2.11)	(3.67)	
Ln Assets (\$ millions)	-0.018	-0.361**	-0.336***	-0.030	0.257	-0.091	
	(-0.99)	(-2.09)	(-2.94)	(-1.14)	(1.17)	(-0.43)	
Expense Ratio	-0.085	-0.750	-1.059 **	-0.251***	-1.106	-2.544***	
	(-1.13)	(-1.06)	(-2.06)	(-2.79)	(-1.27)	(-3.07)	
Turnover	0.000	0.000	-0.002	-0.000	0.005	0.009**	
	(0.63)	(0.04)	(-0.69)	(-0.07)	(1.12)	(2.18)	
Ln Flows (\$ millions)	0.013***	0.081***	0.071***	0.011***	0.081***	0.097***	
	(7.64)	(4.57)	(5.45)	(3.95)	(3.21)	(4.74)	
Style FE	YES	YES	YES	YES	YES	YES	
Time FE	YES	YES	YES	YES	YES	YES	
Observations	1,125	1,125	1,125	480	480	480	
Adj. R-squared	0.431	0.102	0.069	0.486	0.134	0.126	

Table V: Matched Sample Analysis of the Changes in Fund Performance after the Switch

This table reports the comparisons of the risk-adjusted performance of the incumbent and acquired funds to the performance of matched samples of funds over the sample period of January 1980 to December 2012. Panel A contains the results comparing the change in performance of the incumbent funds to that of i) the propensity-score-matched and ii) randomly-matched samples of funds. Panel B contains the results comparing the change in performance of the acquired funds to that of the corresponding matched samples. The propensity score matched sample for the incumbent (acquired) funds is constructed based on the propensity scores estimated from Model 1 (4) of Table 2. The dependent variable in each regression is a measure of risk-adjusted performance. The dependent variable in models (1) and (4) in each panel is the one-year Sharpe ratio measured over the 12-month windows before and after the switch. The dependent variable in models (3) and (6) in each panel is the one-year Morningstar Category-adjusted alpha estimated over the 12-month windows before and after the switch. The dependent variable in models (3) and (6) in each panel is the one-year Morningstar Category-adjusted alpha estimated over the 12-month windows before and after the switch. The main independent variable of interest is *After* that equals one (zero) if the observation is within the 12-month period after (before) the managers' switch to multitasking. Other independent variables include the natural logarithm of the fund's average total net assets (in millions of dollars), the average expense ratio (in %), the average turnover ratio (in %), and the natural logarithm of net dollar flows (in millions of dollars), all measured over the 12-month periods before and after the switch date. The last two rows in each panel report the differences in the coefficient on *After* and the *p*-values from the *F*-tests of the differences. All variables are winsorized at the 1st and the 99th percentile levels. We control for the investment style and year fixed e

	Propensity-Score Matching				Random Matching	2
	(1)	(2)	(3)	(4)	(5)	(6)
	Sharpe	Alpha	MS Alpha	Sharpe	Alpha	MS Alpha
Regression for Multitasking Funds						
After (Multitasking)	-0.170**	-1.662***	-1.261***	-0.172**	-1.894***	-1.580***
·	(-2.02)	(-2.66)	(-3.01)	(-2.15)	(-3.15)	(-3.58)
Regression for Matched Funds				. ,		· · · ·
After (Matched)	-0.066	0.261	0.655	-0.041	0.390	-0.057
	(-0.85)	(0.45)	(1.61)	(-0.57)	(0.74)	(-0.13)
Diff. (Multitasking – Matched)	-0.102	-1.933**	-2.122***	-0.131	-2.284***	-1.523**
<i>p</i> -value (F-Test of Difference)	0.364	0.024	0.001	0.217	0.005	0.015

Panel A: Incumbent Funds

Panel B: Acquired Funds

	Propensity-Score Matching		Random Mat		g	
	(1)	(1) (2)	(3)	(1)	(2)	(3)
	Sharpe	Alpha	MS Alpha	Sharpe	Alpha	MS Alpha
After (Multitasking)	0.199**	1.924**	2.674***	0.188*	1.932**	2.255***
	(2.01)	(2.03)	(3.71)	(1.95)	(2.13)	(3.24)
After (Matched)				-0.018	-0.518	0.167
	0.041	-0.665	0.227	(-0.19)	(-0.57)	(0.24)
	(0.42)	(-0.71)	(0.32)			
Diff. (Multitasking – Matched)				0.206	2.45*	2.088**
<i>p</i> -value (F-Test of Difference)	0.158	2.589*	2.447**	0.149	0.058	0.035

Table VI: Impact of Acquired Fund Characteristics on Incumbent Fund Performance

This table reports regressions examining the impact of the manager's effort diversion on the performance of the incumbent funds after their manager's switch to multitasking. Panel A contains the results when the characteristic considered is New Fund, an indicator variable equal to one (zero) if the fund the manager acquires is new (existing). Panel B contains the results when the characteristic considered is AcquirerTurnover, which is equal to the acquired fund's turnover ratio for the one-year after the manager begins running it. Panel C contains the results when the characteristic considered is Acquired Style Deviation, which is the absolute value of the acquired fund's beta from a single factor regression of its returns on the returns of its Morningstar Style Index minus one for the year after the manager begins running that fund. The dependent variable in model (1) is the one-year Sharpe ratio calculated from the 12-month periods before and after the switch. The dependent variable in model (2) is the one-year Carhart (1997) four-factor alpha estimated from the 12-month periods before and after the switch. The dependent variable in models (3) is the one-year Morningstar Category-adjusted alpha estimated from the 12-month periods before and after the switch. The main independent variable of interest is After $\times Y$, which is the interaction of After, an indicator variable that equals one (zero) if the observation is within the 12-month period after (before) the managers' switch to multitasking, and an acquirer characteristic, Y. Other independent variables include the natural logarithm of the fund's average total net assets (in millions of dollars), the average expense ratio (in %), the average turnover ratio (in %), and the natural logarithm of net dollar flows (in millions of dollars), all estimated or measured over the 12month windows before and after the switch. All variables are winsorized at the 1st and the 99th percentile levels. We control for the investment style and year fixed effects. The standard errors are clustered at the fund level. The tstatistics are reported in the parentheses. Statistical significance of 1%, 5%, and 10% is indicated by ***,**, and * respectively.

	(1)	(2)	(3)
	Sharpe	Alpha	MS Alpha
After × Effort Proxy	-0.234**	-1.066	-2.136***
	(-2.43)	(-1.11)	(-2.90)
After	-0.045	-1.151	0.003
	(-0.52)	(-1.41)	(0.00)
Ln Assets (\$ millions)	-0.018	-0.377**	-0.369***
	(-0.89)	(-2.02)	(-3.12)
Expense Ratio	-0.123	-1.270	-1.356**
	(-1.49)	(-1.62)	(-2.48)
Turnover	0.001*	0.004	-0.001
	(1.80)	(0.77)	(-0.28)
Ln Flows (\$ millions)	0.013***	0.074***	0.063***
	(7.33)	(3.80)	(4.24)
Style FE	YES	YES	YES
Time FE	YES	YES	YES
Observations	1,355	1,355	1,355
Adj. R-squared	0.434	0.115	0.091

Panel A: Effort Proxy = New Fund Dummy

	(1)	(2)	(3)
	Sharpe	Alpha	MS Alpha
After × Effort Proxy	-0.066***	-1.002***	-0.453***
	(-3.37)	(-6.68)	(-4.76)
After	-0.079	-0.697	-0.646
	(-1.07)	(-0.93)	(-1.25)
Ln Assets (\$ millions)	-0.017	-0.404**	-0.420***
	(-0.74)	(-1.98)	(-3.39)
Expense Ratio	-0.075	-1.184	-1.154*
	(-0.86)	(-1.42)	(-1.95)
Turnover	0.001***	0.011**	0.003
	(2.63)	(2.37)	(0.93)
Ln Flows (\$ millions)	0.013***	0.065***	0.059***
	(6.74)	(3.13)	(3.59)
Style FE	YES	YES	YES
Time FE	YES	YES	YES
Observations	1,189	1,189	1,189
Adj. R-squared	0.437	0.132	0.090

Panel B. Effort Proxy = Acquired Fund Turnover

Panel C. Effort Proxy = Acquired Style Deviation

	(1)	(2)	(3)
	Sharpe	Alpha	MS Alpha
After × Effort Proxy	-0.423**	-3.727**	-1.349
	(-2.36)	(-2.11)	(-1.08)
After	-0.056	-1.292	-0.883
	(-0.67)	(-1.52)	(-1.55)
Ln Assets (\$ millions)	-0.034	-0.330	-0.438***
	(-1.55)	(-1.50)	(-3.35)
Expense Ratio	-0.116	-1.439	-0.992
	(-1.21)	(-1.60)	(-1.47)
Turnover	0.001**	0.007	0.001
	(2.30)	(1.17)	(0.39)
Ln Flows (\$ millions)	0.013***	0.075***	0.062***
	(6.74)	(3.44)	(3.77)
Style FE	YES	YES	YES
Time FE	YES	YES	YES
Observations	1.035	1.035	1.035
Adj. R-squared	0.472	0.124	0.100

Table VII: Multivariate Analysis of the Changes in Fund Flows after the Switch

This table reports the results of regressions examining the impact of multitasking on the fund flows of the incumbent funds (models (1) and (2)) and the acquired funds (models (3) and (4)) before (*i.e.*, month t-15 to t-3) and after (*i.e.*, month t+3 to t+15) the managers' switch to multitasking. The dependent variable in models (1) and (3) is the net dollar flows estimated from reported returns and total net assets as in equation (2). The dependent variables are either estimated or aggregated monthly dollar flows from the N-SAR filings. All the dependent variables are either estimated or aggregated over the 12-month periods before and after the switch. The main independent variable of interest is an indicator variable, *After*, that equals one (zero) if the observation is within the 12-month period after (before) the managers' switch to multitasking. Other independent variables include the contemporaneous and lagged one-year Carhart (1997) four-factor alphas (in %), the natural logarithm of the average fund's total net assets (in millions of dollars), average expense ratio (in %), average turnover ratio, and the natural logarithm of fund's lagged net flows. We control for the investment style and year fixed effects. The standard errors are clustered at the fund level. The *t*-statistics are reported in the parentheses. Statistical significance of 1%, 5%, and 10% is indicated by ***,**, and * respectively.

	Incumbent Funds		Acquir	ed Funds
	(1) (2)		(3)	(4)
	Net Flows	NSAR Flows	Net Flows	NSAR Flows
After	-2.956	15.256	22.397**	30.364**
	(-0.21)	(0.58)	(2.45)	(2.04)
Four Factor Alpha	3.102***	3.772***	1.633***	1.813*
-	(3.85)	(2.85)	(2.78)	(1.83)
Lag Four–Factor Alpha	3.349***	1.481	0.998**	-0.393
-	(4.26)	(1.10)	(2.05)	(-0.51)
Ln Assets (\$ millions)	26.329***	-6.196	-22.720***	-28.068***
	(4.15)	(-0.71)	(-5.17)	(-4.12)
Expense Ratio	17.567	-25.274	9.474	8.857
	(1.01)	(-0.88)	(0.69)	(0.39)
Turnover	-0.132*	-0.179**	-0.040	-0.076
	(-1.88)	(-2.28)	(-0.82)	(-0.93)
Ln Flows (\$ millions)	4.729***	5.018***	2.399***	2.224***
	(9.42)	(5.92)	(6.39)	(3.24)
Style FE	YES	YES	YES	YES
Year FE	YES	YES	YES	YES
Observations	1,082	394	465	216
Adj. R-squared	0.157	0.0966	0.257	0.261

Table VIII: Fund Flows into the New Funds: Multitasking versus Single-tasking

This table compares the net dollar flows into the new funds launched by multitasking managers versus those launched by single-tasking managers over the 12-month period after the launch of a new fund. The dependent variable is either the estimated dollar flows (model (1)) or aggregate N-SAR dollar flows (model (2)) as defined in Table VII. The main independent variable of interest is an indicator variable, *Multitasking*, that equals one if a new fund is launched by a manager to multitask, and zero if the new fund is the only fund managed by the manager. Other independent variables include the one-year Carhart (1997) four-factor alpha (in %), the average expense ratio (in %), and average turnover ratio (in %). We control for the investment style and year fixed effects. The standard errors are clustered at the fund level. The *t*-statistics are reported in the parentheses. Statistical significance of 1%, 5%, and 10% is indicated by ***,**, and * respectively.

	(1)	(2)
	Net Flows	NSAR Flows
Multitasking	9.274**	14.526*
	(1.98)	(1.67)
Four–Factor Alpha	0.251	-0.025
	(1.12)	(-0.06)
Expense Ratio	-5.918*	-14.379**
	(-1.94)	(-2.20)
Turnover	0.016	0.018
	(0.96)	(0.55)
Style FE	YES	YES
Year FE	YES	YES
Observations	751	253
Adj. R-squared	0.041	-0.002

Table IX: Determinants and Effects of Switchbacks from Multitasking to Single-Tasking

Panel A of this table reports the results from logistic regressions analyzing the determinants of a manager switching back from multitasking to single-tasking (i.e., switchback). The dependent variable for each model is SwitchBack, an indicator variable equal to one in the month a manager stops multitasking and zero otherwise. The independent variables are the characteristics of the managers' incumbent fund and include a measure its risk-adjusted performance (Sharpe ratio, Carhart (1997) four-factor alpha, or Morningstar Category-adjusted alpha), its average total net assets (in millions of dollars), average expense ratio (in %), average turnover ratio (in %), and the natural logarithm of net dollar flows (in millions of dollars), all estimated or measured over the 12-month period three months prior to the month of the switch. Panel B reports regressions of the risk-adjusted performance of the fund retained by the managers who switch back from multitasking to single-tasking (*i.e.*, switch-back) over the 12-month period before (*i.e.*, month t-15 to t-3) and after (*i.e.*, month t+3 to t+15) the switchback. The dependent variable in model (1) is the one-year Sharpe ratio. The dependent variable in model (2) is the one-year Carhart (1997) fourfactor alpha estimated over the 12-month period. The dependent variable in model (3) is the one-year Morningstar Category-adjusted alpha estimated over the 12-month period. The main independent variable of interest is After, an indicator variable equal to one (zero) if the observation is within the 12-month period after (before) the managers' switch back from multitasking to single-tasking. The control variables are the same as in Panel A. All variables are winsorized at the 1st and the 99th percentile levels. Our sample period is January 1980 to December 2012. We control for investment style and year fixed effects in all regressions. The standard errors are clustered at the fund level. The t-statistics are reported in the parentheses. Statistical significance of 1%, 5%, and 10% is indicated by ***, **, and * respectively.

	(1)	(2)	(3)
Sharpe Ratio	-0.121**		
	(-2.15)		
Four Factor Alpha		-0.016***	
		(-2.96)	
MS Alpha			-0.029***
			(-2.62)
Ln Assets (\$ millions)	-0.132***	-0.132***	-0.132***
	(-3.40)	(-3.41)	(-3.40)
Expense Ratio	-0.122	-0.118	-0.117
	(-0.71)	(-0.70)	(-0.69)
Turnover	0.002***	0.002***	0.002***
	(2.93)	(2.75)	(2.87)
Ln Flows (\$ millions)	-0.011***	-0.011***	-0.011***
	(-3.07)	(-3.02)	(-2.91)
Style FE	YES	YES	YES
Time FE	YES	YES	YES
Observations	35,505	35,505	35,505
Pseudo R ²	0.0325	0.0334	0.0329

Panel A. Determinants of Switchbacks

Panel B. Effects of Switchbacks

	(1)	(2)	(3)
	Sharpe	Alpha	MS Alpha
After	0.154*	1.666**	1.159**
	(1.67)	(2.11)	(1.99)
Ln Assets (\$ millions)	0.018	0.026	0.133
	(0.84)	(0.11)	(0.72)
Expense Ratio	-0.170**	0.453	-0.307
	(-2.08)	(0.49)	(-0.38)
Turnover	0.000	-0.002	0.005
	(0.63)	(-0.43)	(1.19)
Ln Flows (\$ millions)	0.009***	0.071**	0.053**
	(3.45)	(2.59)	(2.52)
Stule FF	VES	VES	VES
Time EE	I LS VES	I LS VES	I ES VES
Time FE	YES	YES	YES
Observations	570	570	570
Adj. R–squared	0.426	0.114	0.063

CFR working paper series



CFR Working Papers are available for download from www.cfr-cologne.de.

Hardcopies can be ordered from: Centre for Financial Research (CFR), Albertus Magnus Platz, 50923 Koeln, Germany.

2015

No.	Author(s)	Title
15-08	V. Agarwal, T. C. Green, H. Ren	Alpha or Beta in the Eye of the Beholder: What drives Hedge Fund Flows?
15-07	V. Agarwal, S. Ruenzi, F. Weigert	Tail risk in hedge funds: A unique view from portfolio holdings
15-06	C. Lan, F. Moneta, R. Wermers	Mutual Fund Investment Horizon and Performance
15-05	L.K. Dahm, C. Sorhage	Milk or Wine: Mutual Funds' (Dis)economies of Life
15-04	A. Kempf, D. Mayston, M. Gehde-Trapp, P. K. Yadav	Resiliency: A Dynamic View of Liquidity
15-03	V. Agarwal, Y. E. Arisoy, N. Y. Naik	Volatility of Aggregate Volatility and Hedge Funds Returns
15-02	G. Cici, S. Jaspersen, A. Kempf	Speed of Information Diffusion within Fund Families
15-01	M. Baltzer, S. Jank, E. Smajlbegovic	Who trades on momentum?

No.	Author(s)	Title
14-14	G. Cici, L. K. Dahm, A. Kempf	Trading Efficiency of Fund Families: Impact on Fund Performance and Investment Behavior
14-13	V. Agarwal, Y. Lu, S. Ray	Under one roof: A study of simultaneously managed hedge funds and funds of hedge funds
14-12	P. Limbach, F. Sonnenburg	CEO Fitness and Firm Value
14-11	G. Cici, M. Gehde-Trapp, M. Göricke, A. Kempf	What They Did in their Previous Life: The Investment Value of Mutual Fund Managers' Experience outside the Financial Sector

14-10	O. Korn, P. Krischak, E. Theissen	Illiquidity Transmission from Spot to Futures Markets
14-09	E. Theissen, L. S. Zehnder	Estimation of Trading Costs: Trade Indicator Models Revisited
14-08	C. Fink, E. Theissen	Dividend Taxation and DAX Futures Prices
14-07	F. Brinkmann, O. Korn	Risk-adjusted Option-implied Moments
14-06	J. Grammig, J. Sönksen	Consumption-Based Asset Pricing with Rare Disaster Risk
14-05	J. Grammig, E. Schaub	Give me strong moments and time – Combining GMM and SMM to estimate long-run risk asset pricing
14-04	C. Sorhage	Outsourcing of Mutual Funds' Non-core Competencies
14-03	D. Hess, P. Immenkötter	How Much Is Too Much? Debt Capacity And Financial Flexibility
14-02	C. Andres, M. Doumet, E. Fernau, E. Theissen	The Lintner model revisited: Dividends versus total payouts
14-01	N.F. Carline, S. C. Linn, P. K. Yadav	Corporate Governance and the Nature of Takeover Resistance

No.	Author(s)	Title
13-11	R. Baule, O. Korn, S. Saßning	Which Beta is Best? On the Information Content of Option-implied Betas
13-10	V. Agarwal, L. Ma, K. Mullally	Managerial Multitasking in the Mutual Fund Industry
13-09	M. J. Kamstra, L.A. Kramer, M.D. Levi, R. Wermers	Seasonal Asset Allocation: Evidence from Mutual Fund Flows
13-08	F. Brinkmann, A. Kempf, O. Korn	Forward-Looking Measures of Higher-Order Dependencies with an Application to Portfolio Selection
13-07	G. Cici, S. Gibson, Y. Gunduz, J.J. Merrick, Jr.	Market Transparency and the Marking Precision of Bond Mutual Fund Managers
13-06	S. Bethke, A. Kempf, M. Trapp	Investor Sentiment, Flight-to-Quality, and Corporate Bond Comovement
13-05	P. Schuster, M. Trapp, M. Uhrig-Homburg	A Heterogeneous Agents Equilibrium Model for the Term Structure of Bond Market Liquidity
13-04	V. Agarwal, K. Mullally, Y. Tang, B. Yang	Mandatory Portfolio Disclosure, Stock Liquidity, and Mutual Fund Performance
13-03	V. Agarwal, V. Nanda, S.Ray	Institutional Investment and Intermediation in the Hedge Fund Industry
13-02	C. Andres, A. Betzer, M. Doumet, E. Theissen	Open Market Share Repurchases in Germany: A Conditional Event Study Approach
13-01	J. Gaul, E. Theissen	A Partially Linear Approach to Modelling the Dynamics of Spot and Futures Price

No.	Author(s)	Title
12-12	Y. Gündüz, J. Nasev, M. Trapp	The Price Impact of CDS Trading
12-11	Y. Wu, R. Wermers, J. Zechner	Governance and Shareholder Value in Delegated Portfolio Management: The Case of Closed-End Funds
12-10	M. Trapp, C. Wewel	Transatlantic Systemic Risk
12-09	G. Cici, A. Kempf, C. Sorhage	Do Financial Advisors Provide Tangible Benefits for Investors? Evidence from Tax-Motivated Mutual Fund Flows
12-08	S. Jank	Changes in the composition of publicly traded firms: Implications for the dividend-price ratio and return predictability
12-07	G. Cici, C. Rosenfeld	The Investment Abilities of Mutual Fund Buy-Side Analysts
12-06	A. Kempf, A. Pütz, F. Sonnenburg	Fund Manager Duality: Impact on Performance and Investment Behavior
12-05	R. Wermers	Runs on Money Market Mutual Funds
12-04	R. Wermers	A matter of style: The causes and consequences of style drift in institutional portfolios
12-02	C. Andres, E. Fernau, E. Theissen	Should I Stay or Should I Go? Former CEOs as Monitors
12-01	L. Andreu, A. Pütz	Are Two Business Degrees Better Than One? Evidence from Mutual Fund Managers' Education

No.	Author(s)	Title
11-16	V. Agarwal, JP. Gómez, R. Priestley	Management Compensation and Market Timing under Portfolio Constraints
11-15	T. Dimpfl, S. Jank	Can Internet Search Queries Help to Predict Stock Market Volatility?
11-14	P. Gomber, U. Schweickert, E. Theissen	Liquidity Dynamics in an Electronic Open Limit Order Book: An Event Study Approach
11-13	D. Hess, S. Orbe	Irrationality or Efficiency of Macroeconomic Survey Forecasts? Implications from the Anchoring Bias Test
11-12	D. Hess, P. Immenkötter	Optimal Leverage, its Benefits, and the Business Cycle
11-11	N. Heinrichs, D. Hess, C. Homburg, M. Lorenz, S. Sievers	Extended Dividend, Cash Flow and Residual Income Valuation Models – Accounting for Deviations from Ideal Conditions
11-10	A. Kempf, O. Korn, S. Saßning	Portfolio Optimization using Forward - Looking Information
11-09	V. Agarwal, S. Ray	Determinants and Implications of Fee Changes in the Hedge Fund Industry
11-08	G. Cici, LF. Palacios	On the Use of Options by Mutual Funds: Do They Know What They Are Doing?

11-07	V. Agarwal, G. D. Gay, L. Ling	Performance inconsistency in mutual funds: An investigation of window-dressing behavior
11-06	N. Hautsch, D. Hess, D. Veredas	The Impact of Macroeconomic News on Quote Adjustments, Noise, and Informational Volatility
11-05	G. Cici	The Prevalence of the Disposition Effect in Mutual Funds' Trades
11-04	S. Jank	Mutual Fund Flows, Expected Returns and the Real Economy
11-03	G.Fellner, E.Theissen	Short Sale Constraints, Divergence of Opinion and Asset Value: Evidence from the Laboratory
11-02	S.Jank	Are There Disadvantaged Clienteles in Mutual Funds?
11-01	V. Agarwal, C. Meneghetti	The Role of Hedge Funds as Primary Lenders

No.	Author(s)	Title
10-20	G. Cici, S. Gibson, J.J. Merrick Jr.	Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds
10-19	J. Hengelbrock, E. Theissen, C. Westheide	Market Response to Investor Sentiment
10-18	G. Cici, S. Gibson	The Performance of Corporate-Bond Mutual Funds: Evidence Based on Security-Level Holdings
10-17	D. Hess, D. Kreutzmann, O. Pucker	Projected Earnings Accuracy and the Profitability of Stock Recommendations
10-16	S. Jank, M. Wedow	Sturm und Drang in Money Market Funds: When Money Market Funds Cease to Be Narrow
10-15	G. Cici, A. Kempf, A. Puetz	The Valuation of Hedge Funds' Equity Positions
10-14	J. Grammig, S. Jank	Creative Destruction and Asset Prices
10-13	S. Jank, M. Wedow	Purchase and Redemption Decisions of Mutual Fund Investors and the Role of Fund Families
10-12	S. Artmann, P. Finter, A. Kempf, S. Koch, E. Theissen	The Cross-Section of German Stock Returns: New Data and New Evidence
10-11	M. Chesney, A. Kempf	The Value of Tradeability
10-10	S. Frey, P. Herbst	The Influence of Buy-side Analysts on Mutual Fund Trading
10-09	V. Agarwal, W. Jiang, Y. Tang, B. Yang	Uncovering Hedge Fund Skill from the Portfolio Holdings They Hide
10-08	V. Agarwal, V. Fos, W. Jiang	Inferring Reporting Biases in Hedge Fund Databases from Hedge Fund Equity Holdings
10-07	V. Agarwal, G. Bakshi, J. Huij	Do Higher-Moment Equity Risks Explain Hedge Fund Returns?
10-06	J. Grammig, F. J. Peter	Tell-Tale Tails: A data driven approach to estimate unique market information shares

10-05	K. Drachter, A. Kempf	Höhe, Struktur und Determinanten der Managervergütung- Eine Analyse der Fondsbranche in Deutschland
10-04	J. Fang, A. Kempf, M. Trapp	Fund Manager Allocation
10-03	P. Finter, A. Niessen- Ruenzi, S. Ruenzi	The Impact of Investor Sentiment on the German Stock Market
10-02	D. Hunter, E. Kandel, S. Kandel, R. Wermers	Mutual Fund Performance Evaluation with Active Peer Benchmarks
10-01	S. Artmann, P. Finter, A. Kempf	Determinants of Expected Stock Returns: Large Sample Evidence from the German Market

R. Wermers

No.	Author(s)	Title
09-17	E. Theissen	Price Discovery in Spot and Futures Markets: A Reconsideration
09-16	М. Тгарр	Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity
09-15	A. Betzer, J. Gider, D.Metzger, E. Theissen	Strategic Trading and Trade Reporting by Corporate Insiders
09-14	A. Kempf, O. Korn, M. Uhrig-Homburg	The Term Structure of Illiquidity Premia
09-13	W. Bühler, M. Trapp	Time-Varying Credit Risk and Liquidity Premia in Bond and CDS Markets
09-12	W. Bühler, M. Trapp	Explaining the Bond-CDS Basis – The Role of Credit Risk and Liquidity
09-11	S. J. Taylor, P. K. Yadav, Y. Zhang	Cross-sectional analysis of risk-neutral skewness
09-10	A. Kempf, C. Merkle, A. Niessen-Ruenzi	Low Risk and High Return – Affective Attitudes and Stock Market Expectations
09-09	V. Fotak, V. Raman, P. K. Yadav	Naked Short Selling: The Emperor`s New Clothes?
09-08	F. Bardong, S.M. Bartram, P.K. Yadav	Informed Trading, Information Asymmetry and Pricing of Information Risk: Empirical Evidence from the NYSE
09-07	S. J. Taylor , P. K. Yadav, Y. Zhang	The information content of implied volatilities and model-free volatility expectations: Evidence from options written on individual stocks
09-06	S. Frey, P. Sandas	The Impact of Iceberg Orders in Limit Order Books
09-05	H. Beltran-Lopez, P. Giot, J. Grammig	Commonalities in the Order Book
09-04	J. Fang, S. Ruenzi	Rapid Trading bei deutschen Aktienfonds: Evidenz aus einer großen deutschen Fondsgesellschaft
09-03	A. Banegas, B. Gillen, A. Timmermann,	The Cross-Section of Conditional Mutual Fund Performance in European Stock Markets

09-02	J. Grammig, A. Schrimpf, M. Schuppli	Long-Horizon Consumption Risk and the Cross-Section of Returns: New Tests and International Evidence
09-01	O. Korn, P. Koziol	The Term Structure of Currency Hedge Ratios

_	No.	Author(s)	Title
	08-12	U. Bonenkamp, C. Homburg, A. Kempf	Fundamental Information in Technical Trading Strategies
	08-11	O. Korn	Risk Management with Default-risky Forwards
	08-10	J. Grammig, F.J. Peter	International Price Discovery in the Presence of Market Microstructure Effects
	08-09	C. M. Kuhnen, A. Niessen	Public Opinion and Executive Compensation
	08-08	A. Pütz, S. Ruenzi	Overconfidence among Professional Investors: Evidence from Mutual Fund Managers
	08-07	P. Osthoff	What matters to SRI investors?
	08-06	A. Betzer, E. Theissen	Sooner Or Later: Delays in Trade Reporting by Corporate Insiders
	08-05	P. Linge, E. Theissen	Determinanten der Aktionärspräsenz auf Hauptversammlungen deutscher Aktiengesellschaften
	08-04	N. Hautsch, D. Hess, C. Müller	Price Adjustment to News with Uncertain Precision
	08-03	D. Hess, H. Huang, A. Niessen	How Do Commodity Futures Respond to Macroeconomic News?
	08-02	R. Chakrabarti, W. Megginson, P. Yadav	Corporate Governance in India
	08-01	C. Andres, E. Theissen	Setting a Fox to Keep the Geese - Does the Comply-or-Explain Principle Work?

No.	Author(s)	Title
07-16	M. Bär, A. Niessen, S. Ruenzi	The Impact of Work Group Diversity on Performance: Large Sample Evidence from the Mutual Fund Industry
07-15	A. Niessen, S. Ruenzi	Political Connectedness and Firm Performance: Evidence From Germany
07-14	O. Korn	Hedging Price Risk when Payment Dates are Uncertain
07-13	A. Kempf, P. Osthoff	SRI Funds: Nomen est Omen
07-12	J. Grammig, E. Theissen, O. Wuensche	Time and Price Impact of a Trade: A Structural Approach
07-11	V. Agarwal, J. R. Kale	On the Relative Performance of Multi-Strategy and Funds of Hedge Funds
07-10	M. Kasch-Haroutounian, E. Theissen	Competition Between Exchanges: Euronext versus Xetra
07-09	V. Agarwal, N. D. Daniel, N. Y. Naik	Do hedge funds manage their reported returns?

07-08	N. C. Brown, K. D. Wei, R. Wermers	Analyst Recommendations, Mutual Fund Herding, and Overreaction in Stock Prices
07-07	A. Betzer, E. Theissen	Insider Trading and Corporate Governance: The Case of Germany
07-06	V. Agarwal, L. Wang	Transaction Costs and Value Premium
07-05	J. Grammig, A. Schrimpf	Asset Pricing with a Reference Level of Consumption: New Evidence from the Cross-Section of Stock Returns
07-04	V. Agarwal, N.M. Boyson, N.Y. Naik	Hedge Funds for retail investors? An examination of hedged mutual funds
07-03	D. Hess, A. Niessen	The Early News Catches the Attention: On the Relative Price Impact of Similar Economic Indicators
07-02	A. Kempf, S. Ruenzi, T. Thiele	Employment Risk, Compensation Incentives and Managerial Risk Taking - Evidence from the Mutual Fund Industry -
07-01	M. Hagemeister, A. Kempf	CAPM und erwartete Renditen: Eine Untersuchung auf Basis der Erwartung von Marktteilnehmern

No.	Author(s)	Title
06-13	S. Čeljo-Hörhager, A. Niessen	How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study
06-12	R. Wermers, Y. Wu, J. Zechner	Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers
06-11	U. v. Lilienfeld-Toal, S. Ruenzi	Why Managers Hold Shares of Their Firm: An Empirical Analysis
06-10	A. Kempf, P. Osthoff	The Effect of Socially Responsible Investing on Portfolio Performance
06-09	R. Wermers, T. Yao, J. Zhao	Extracting Stock Selection Information from Mutual Fund holdings: An Efficient Aggregation Approach
06-08	M. Hoffmann, B. Kempa	The Poole Analysis in the New Open Economy Macroeconomic Framework
06-07	K. Drachter, A. Kempf, M. Wagner	Decision Processes in German Mutual Fund Companies: Evidence from a Telephone Survey
06-06	J.P. Krahnen, F.A. Schmid, E. Theissen	Investment Performance and Market Share: A Study of the German Mutual Fund Industry
06-05	S. Ber, S. Ruenzi	On the Usability of Synthetic Measures of Mutual Fund Net- Flows
06-04	A. Kempf, D. Mayston	Liquidity Commonality Beyond Best Prices
06-03	O. Korn, C. Koziol	Bond Portfolio Optimization: A Risk-Return Approach
06-02	O. Scaillet, L. Barras, R. Wermers	False Discoveries in Mutual Fund Performance: Measuring Luck in Estimated Alphas
06-01	A. Niessen, S. Ruenzi	Sex Matters: Gender Differences in a Professional Setting

No.	Author(s)	Title
05-16	E. Theissen	An Analysis of Private Investors' Stock Market Return Forecasts
05-15	T. Foucault, S. Moinas, E. Theissen	Does Anonymity Matter in Electronic Limit Order Markets
05-14	R. Kosowski, A. Timmermann, R. Wermers, H. White	Can Mutual Fund "Stars" Really Pick Stocks? New Evidence from a Bootstrap Analysis
05-13	D. Avramov, R. Wermers	Investing in Mutual Funds when Returns are Predictable
05-12	K. Griese, A. Kempf	Liquiditätsdynamik am deutschen Aktienmarkt
05-11	S. Ber, A. Kempf, S. Ruenzi	Determinanten der Mittelzuflüsse bei deutschen Aktienfonds
05-10	M. Bär, A. Kempf, S. Ruenzi	Is a Team Different From the Sum of Its Parts? Evidence from Mutual Fund Managers
05-09	M. Hoffmann	Saving, Investment and the Net Foreign Asset Position
05-08	S. Ruenzi	Mutual Fund Growth in Standard and Specialist Market Segments
05-07	A. Kempf, S. Ruenzi	Status Quo Bias and the Number of Alternatives - An Empirical Illustration from the Mutual Fund Industry
05-06	J. Grammig, E. Theissen	Is Best Really Better? Internalization of Orders in an Open Limit Order Book
05-05	H. Beltran-Lopez, J. Grammig, A.J. Menkveld	Limit order books and trade informativeness
05-04	M. Hoffmann	Compensating Wages under different Exchange rate Regimes
05-03	M. Hoffmann	Fixed versus Flexible Exchange Rates: Evidence from Developing Countries
05-02	A. Kempf, C. Memmel	Estimating the Global Minimum Variance Portfolio
05-01	S. Frey, J. Grammig	Liquidity supply and adverse selection in a pure limit order book market

No.	Author(s)	Title
04-10	N. Hautsch, D. Hess	Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery
04-09	A. Kempf, K. Kreuzberg	Portfolio Disclosure, Portfolio Selection and Mutual Fund Performance Evaluation
04-08	N.F. Carline, S.C. Linn, P.K. Yadav	Operating performance changes associated with corporate mergers and the role of corporate governance
04-07	J.J. Merrick, Jr., N.Y. Naik, P.K. Yadav	Strategic Trading Behaviour and Price Distortion in a Manipulated Market: Anatomy of a Squeeze
04-06	N.Y. Naik, P.K. Yadav	Trading Costs of Public Investors with Obligatory and Voluntary Market-Making: Evidence from Market Reforms

04-05	A. Kempf, S. Ruenzi	Family Matters: Rankings Within Fund Families and Fund Inflows
04-04	V. Agarwal, N.D. Daniel, N.Y. Naik	Role of Managerial Incentives and Discretion in Hedge Fund Performance
04-03	V. Agarwal, W.H. Fung, J.C. Loon, N.Y. Naik	Risk and Return in Convertible Arbitrage: Evidence from the Convertible Bond Market
04-02	A. Kempf, S. Ruenzi	Tournaments in Mutual Fund Families
04-01	I. Chowdhury, M. Hoffmann, A. Schabert	Inflation Dynamics and the Cost Channel of Monetary Transmission

centre for Financial Research

cfr/university of cologne Albertus-Magnus-Platz D-50923 cologne Fon +49[0]221-470-6995 Fax +49[0]221-470-3992 Kempf@cfr-cologne.de WWW.cfr-cologne.de